CENWP-OD-B                                                                                                                                  01 February 2009
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 5 of 2009, which covers the period from 25 through 31 January 2009.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 135.5 to 197.3 kcfs.  Daily average project tailwater ranged from 12.9’ to 16.6’ msl.  Daily average powerhouse forebay elevation ranged from 72.6’ to 74.9’ msl.  Daily average river temperature ranged from 37ºF to 38ºF.  Secchi disk measurements were between 3.0’ to 5.0’.
1.2 Spill Bay 18, which provides attraction flow for B-Branch, is open 6” during daylight hours.  Daily average spill ranged from 0.5 to 0.6 kcfs. 
1.3 Unit Operation.  Powerhouse 1 remains the priority powerhouse.
1.3.1 Unit 0 was on standby from 0741 until 1158 on 27 January to float trash.

1.3.2 Unit 1 was out of service from 0208 on 30 January until 1517 on 31 January for Neutral Current Transformer replacement.
1.3.3 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.4 Unit 8 has remained out of service since 17 December due to generator lower guide bearing failure.
1.3.5 Unit 9 has remained out of service since 04 August for turbine rehab.
1.3.6 Unit 10 was out of service from 0752 until 1051 on 26 January for trashrack raking.

1.3.7 Unit 11 has remained out of service since 12 June due to rotor damage.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris.
1.4.1 F1 and F2 remain out of service while the WS fishway is dewatered.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week:
2.1.1 FV1-1 was closed from 1551 until 1626 on 26 January for cleaning.

2.1.2 FV3-7 was closed from 1527 until 1541 on 26 January for cleaning.

2.1.3 FV5-3, FV5-4, FV5-9 and FV6-9 remain closed while Cascades Island and WS remain dewatered.
2.2 STS/VBS inspections.  No STS/VBS activities occurred this reporting week.  
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	1, 3
	1/7
	NA
	Removed until 15 September.

	11-18
	NA
	NA
	Six STSs have broken gearboxes due to freezing in December.


2.3 Dewatering and Fish Salvage.  The following fish salvage and dewatering activities occurred this week:
2.3.1 On 27 January Fisheries removed two adult clipped steelhead, one juvenile steelhead and one juvenile chinook from the AFF valve 14 drain chamber.  Fish were released back to the river.
3 RESEARCH  
3.1 WDFW Fish Counting.  Video counting at Bradford Island continues.
3.2 WDFW/ODFW/CRITFC Sea lion hazing.  Hazing from boats continues.
4 FISHWAYS 

4.1 Project biologists inspected 26-27, 29-31January.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND 

	1/27
	A-Branch
	FG3-5 closed
	Under repair
	No action taken
	

	1/27
	B-Branch 
	FG3-24 closed
	Under repair
	No action taken
	

	1/29
	B-Branch
	FG3-24 open
	System in manual
	Operators adjusted
	

	1/29
	B-Branch ent.
	Differential >2.0’
	System in manual
	Operators adjusted.
	

	CASCADES ISLAND- OOS for winter maintenance

	WASHINGTON SHORE- OOS for winter maintenance

	DSM1/ITS – DSM1 is OOS until early September

	DSM2/ JMF – OOS for winter maintenance


4.2.1 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 09/10.  

4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November
4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration was checked on 30 January.  B-branch PLC remains in manual.
4.5.2 WG 2 remains in manual due to an inability to raise >10.9’.  
4.5.3 Cascades Island remains dewatered. 
4.5.4 WS fishway remains dewatered.  On 26 January, divers attempted to repair diffuser B-5 in the south monolith.  There was nothing to repair along the shaft or gate.  The problem appears to be in the motor.  On 27 January, divers located the diffuser with the displaced grating in the collection channel.
4.5.5 The AFF remains dewatered. 
4.5.6 All Bradford Island SLEDs were installed by 30 January.  On 31 January, the SLED at WG1 was lifted so a trapped sea lion could exit the fishway.
4.6 Juvenile Fishways.

4.6.1 DSM1 remains dewatered.
4.6.2 DSM2 remains dewatered. 
4.7 Lamprey Fishways.  BI and WS AWS lamprey ramps remain winterized.
5 CONSTRUCTION
5.1 PH1 Ice and Trash Sluiceway Wall Removal.  Automated gate installation in FY09.  Wall removal in FY10.
5.2 TIE crane.  The TIE crane boom repair contract has been awarded.  Completion expected in Spring ‘09.
6 HAZMAT, SPILLS AND CLEANUP.   Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A 06 November teletype set the BON upper forebay limit as 76.5’ msl. and the lower limit as 71.5’ msl.  

7.2 A 31 December teletype, regarding BON tailwater operations for chum egg incubation, requests a minimum tailwater elevation of 11.5’ msl during all hours from 0001 on 1 January 2009 until further notice. 
JAMES R. MAHAR, P.E.

Operations Manager
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